Title: The importance of central corneal thickness measurements and decision making in general ophthalmology clinics: a masked observational study

Authors:

Ashish A Patwardhan (apatwa@hotmail.com)
Mohammad I Khan (drmikhan@gmail.com)
Susan P Mollan (soozmollan@doctors.org.uk)
Paul M Haigh (paul.haigh@rsh.nhs.uk)

Version: 2 Date: 28 June 2007

Author's response to reviews:

Mr Ashish Patwardhan
4 Quinton Close
Shrewsbury SY3 5PJ
UK
26/5/07
To
The Editor
BMC Ophthalmology
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB
UK
Dear Sir/Madam

I would be grateful if you consider publishing the attached manuscript in your journal. This study highlights the importance of routine use of central corneal thickness measurement in patients with glaucoma. We have confirmed that a pachymeter is an essential item of the ophthalmic equipment armamentarium and it will significantly change patient management in the general ophthalmologist's practice.

Yours sincerely

Ashish Patwardhan